Author’s response to reviews

Title: Psychological distress and anxiety compared amongst dental patients - results of a cross-sectional study in 1549 adults

Authors:

Alexander Zinke (alexander_zinke@posteo.de)

Christian Hannig (Christian.Hannig@uniklinikum-dresden.de)

Hendrik Berth (h.berth@ukdd.de)

Version: 1 Date: 03 Jul 2018

Author’s response to reviews:

The section 'Ethics Approval and Consent to Participate' were changed: "The present study was approved by the Ethics Committee of the Technische Universität Dresden, Germany (EK232062011), and only individuals who provided written informed consent were included as study participants."